Author's response to reviews

Title: A new approach to child mental health care within general practice. An Evaluation study

Authors:

- Peter F.M. Verhaak (p.verhaak@nivel.nl)
- Marloes Van Dijk (marloesvandijk1109@gmail.com)
- Dick Walstock (walstock@xs4all.nl)
- Marieke Zwaanswijk (m.zwaanswijk@nivel.nl)

Version: 2 Date: 25 February 2015

Author's response to reviews:

In this new version of the MS e-mail addresses have been added to the title page. Page- and linenumbering have been added.
A list of abbreviations has been added
Acknowledgments have been added